Author’s response to reviews

Title: Human Papillomavirus (HPV) infection in pregnant women and mother-to-child transmission of genital HPV genotypes: a prospective study in Spain

Authors:

Xavier Castellsagué (xcastellsague@iconcologia.net)
Teresa Drudis (tdrudis@gmail.com)
Maria Paz Cañadas (pazjordi@gmail.com)
Anna Goncé (AGONCE@clinic.ub.es)
Ramón Ros (rros@hmartorell.es)
José M. Pérez (jmperez@clinic.ub.es)
M. Jesús Quintana (MJQuintana@santpau.es)
Jesús Muñoz (jesus@iconcologia.net)
Ginesa Albero (g.albero@iconcologia.net)
Silvia de Sanjosé (s.sanjose@iconcologia.net)
F. Xavier Bosch (x.bosch@iconcologia.net)

Version: 8 Date: 19 February 2009

Author’s response to reviews: see over
Dear Editors of *BMC Infectious Diseases*,

I am pleased to submit revision of manuscript referred above. We have provided the changes made as detailed below.

We really appreciate all the comments and suggestions received from the reviewers which all together have greatly improved the quality of the manuscript.

Sincerely,

Xavier Castellsagué, MD, MPH, Ph.D.
Institut Català d’Oncologia (ICO)
Barcelona, Catalonia, Spain
Changes made:

We have deleted “Multiple types were detected in two infants” in “HPV positivity in infants”

We have changed “p=0.01” by “p=0.02” in “Determinants of HPV infection in children” and “DISCUSSION AND CONCLUSIONS”

We have changed “P=0.02” by “p=0.02” in “Mother to child HPV transmission”

We have changed “[9-13 and reviews [2,3,6-8]” by “([9-13] and reviews [2,3,6-8])” in “DISCUSSION AND CONCLUSIONS”

We have changed “in pregnant women (Table 4).” by “in pregnant women (Table 3).” in “DISCUSSION AND CONCLUSIONS”

We have changed “As shown in Table 5” by “As shown in Table 4” in “DISCUSSION AND CONCLUSIONS”

We have changed “6/25” by “6/25” in Table 4

We have changed “X/X X/31” by “X/X; X/31” in Table 5